Author’s response to reviews

Title: A systematic exploration of differences in contextual factors related to implementing the MOVE! Weight Management Program in VA: A Mixed Methods Study

Authors:

Laura J Damschroder (laura.damschroder@va.gov)
David E Goodrich (david.goodrich2@va.gov)
Claire H Robinson (claire.robinson@va.gov)
Carol E Fletcher (carol.fletcher@va.gov)
Julie C Lowery (julie.lowery@va.gov)

Version: 4 Date: 17 August 2011

Author’s response to reviews:

To: Editor, BMC Health Services Research
FROM: Laura J. Damschroder, Corresponding Author

We have made the changes suggested by Reviewer 1 in this round of reviews with “track changes” to enable clearly, the changes made.

I emailed yesterday about needing to know the US Tax ID for BioMed Central in order to pay the publishing fee. We have found this entity in the US VA database of vendors but the record is missing the Tax ID which is needed. Is one available? We are fast approaching the end of our fiscal year and need pay this fee as soon as is possible, to avoid delays resulting from our onerous budgeting process for the new fiscal year.

We have reformatted the tables to comply with formatting standards. However, the amile specified that you are unable to display “…text within tables.” Our tables do contain text. Is this an issue? If so, please let me know and I’ll use images of the tables instead (is this an option?). The figures have been converted to bmp files to enable cropping as requested. We have also replaced the table in Additional File 1 with the Master Interview Guide which includes the quantitative items with a citation for source.

This paper complies with the RATS guidelines for qualitative studies. If you have any questions or concerns, please do not hesitate to contact me.

Sincerely,
Laura Damschroder